Author's response to reviews

Title: Cross-cultural adaptation of the VISA-A questionnaire, an index of clinical severity for patients with Achilles tendinopathy, with reliability, validity and structure evaluations

Authors:

Karin Gravare Silbernagel (karin.gravare-silbernagel@orthop.gu.se)
Roland Thomee (roland.thomee@orthop.gu.se)
Jon Karlsson (jon.karlsson@vgregion.se)

Version: 3 Date: 4 March 2005

Author’s response to reviews: see over
Göteborg March 4th 2005

BMC Musculoskeletal Disorders

Re: MS: 3160655025226095 – Cross-cultural adaptation of the VISA-A questionnaire, an index of clinical severity for patients with Achilles tendinopathy, with reliability, validity and structure evaluations

We have completed the revision of our manuscript and amended all the formatting requested below.

Author list - Please remove titles/qualifications from the list of authors.  
Done

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals).  
Done

Throughout the abstract file and manuscript text please replace VISA Questionnaire with VISA questionnaire.  
Done

References - The reference list should be formatted in the journal's style to enable us to create links to article abstracts on PubMed, please can you re-format the authors names in the reference list as shown below:  
Done

Sincerely,

Karin Grävare Silbernagel, PT, MSc

Lundberg Laboratory for Orthopaedic Research, Dept of Orthopaedics, Göteborg University,
Sahlgrenska University Hospital, Gröna Stråket 12, SE-413 45 Göteborg, Sweden
Fax: +46 31 416816 e-mail: karin.gravare-silbernagel@orthop.gu.se